Author’s response to reviews

Title: How evidence-based is an 'evidence-based parenting programme'? A PRISMA systematic review and meta-analysis of Triple P.

Authors:

Philip Wilson (p.wilson@abdn.ac.uk)
Robert Rush (rrush@gmu.ac.uk)
Susan Hussey (susan.hussey@nhs.net)
Christine Puckering (christine.puckering@glasgow.ac.uk)
Fiona Sim (fiona.sim@glasgow.ac.uk)
Clare S Allely (clare.alley@glasgow.ac.uk)
Paul Doku (dokupaul@yahoo.co.uk)
Alex McConnachie (alex.mcconnachie@glasgow.ac.uk)
Christopher Gillberg (christopher.gillberg@gnc.gu.se)

Version: 5 Date: 10 October 2012

Author’s response to reviews:

Dear Dr Alam,

Thank you for giving provisional acceptance to our paper and thanks once again for providing helpful feedback. We have made three changes to the manuscript in line with your comment and the suggestions of Referee 2:

- The penultimate sentence on page 10 has been altered slightly to indicate the difficulty that we had in establishing author affiliation in some cases
- The second sentence of the final paragraph on page 17 has been altered for clarity
- The sentence in the Acknowledgements section thanking our referees has been deleted.

We hope you now find the paper suitable for publication and we look forward to hearing from you.

With best wishes

Philip Wilson, on behalf of the authors